Author's response to reviews

Title: Association of C-reactive protein, tumor necrosis factor-alpha, and interleukin-6 with chronic kidney disease

Authors:

Belinda T Lee (blee14@tulane.edu)
Faheemuddin A Ahmed (drgaheemuddin@gmail.com)
L. Lee Hamm (lhamm@tulane.edu)
Federico J Teran (fteran@tulane.edu)
Chung-Shiuan Chen (cchen1@tulane.edu)
Yanxi Liu (liuyanxi9@gmail.com)
Kamal Shah (KShah@tulane.edu)
Nader Rifai (Nader.Rifai@childrens.harvard.edu)
Vecihi Batuman (vbatuma@tulane.edu)
Eric E Simon (esimon@tulane.edu)
Jiang He (jhe@tulane.edu)
Jing Chen (jchen@tulane.edu)

Version: 2
Date: 21 November 2014

Author's response to reviews: see over
November 21, 2014

Dr. Hayley Henderson
Executive Editor Hayley Henderson
BMC Nephrology

RE: Association of C-reactive protein, tumor necrosis factor-alpha, and interleukin-6 with chronic kidney disease

Dear Dr. Henderson,

Attached please find a revised manuscript entitled “Association of C-reactive protein, tumor necrosis factor-alpha, and interleukin-6 with chronic kidney disease” for your review.

Thank you very much for your comments and suggestions. We have revised our manuscript by adding more detailed information on each author and including line numbers.

Thank you for the opportunity to submit our work.

Sincerely,

Jing Chen, MD, MSc
Associate Professor of Medicine
Tulane University School of Medicine